Author's response to reviews

Title: Nebulised Hypertonic Saline (3%) Among Children With Mild To Moderately Severe Bronchiolitis - A Double Blind Randomized Controlled Trial.

Authors:

Aayush Khanal Teaching Assistant (aayush_khanal@yahoo.com)
Arun K Sharma Associate Professor (docarunsharma@hotmail.com)
Srijana Basnet Assistant Professor (dr.srijanabasnet@yahoo.com)
Pushpa R Sharma Professor (pushparaj.sharma@gmail.com)
Fakir C Gami Professor (fakirgami@hotmail.com)

Date: 23 April 2015

Author's response to reviews:

To,
The Handling Editor,
BMC Pediatrics,

Respected Sir/ Ma’am,

Most humbly and respectfully, i’d like to state that i’ve made all the necessary changes as suggested by the editorial feedbacks. I’ve mentioned below the changes made:

1. Editorial Feedbacks were ‘Table 1: RR 60 should be >60’. The suggested revision has been made and most right column now reads RR >60.

2. Editorial Feedbacks were 'Table 2 is missing'. The Table-2. Baseline Characteristics.doc has been uploaded.

3. Editorial Feedbacks were ‘Table 4: Both the old and the revised version are shown; please, delete the old version. In the revised version, I guess that -0.37; -0.25 (-0.06) should be -0.37; 0.25 (-0.06).” The suggested revision has been made as -0.37; 0.25 (-0.06).

I’d like to express my gratefulness for the editorial team for highlighting these subtle yet significant errors in the manuscript and for such swift processing of the manuscript.

Sincerely yours,

Dr. Aayush Khanal, MD